

https://doi.org/10.1093/pnasnexus/pgad247 Advance access publication 1 August 2023 Research Report

Maternal-derived galectin-1 shapes the placenta niche through Sda terminal glycosylation: Implication for preeclampsia

Yiran Xie D^a, Fangqi Zhao D^{a,1}, Nancy Freitag^{a,b,1}, Sophia Borowski D^{a,b,1,2}, Yiru Wang D^a, Charlotte Harms D^a, Poh-Choo Pang^c, Juliette Desforges^c, Tianyu Wen D^c, Edzard Schwedhelm D^d, Manvendra Singh^e, Ralf Dechend^{b,f}, Anne Dell D^c, Stuart M. Haslam^c, Gabriela Dveksler D^g, Mariana G. Garcia D^a and Sandra M. Blois D^{a,*}

^aDepartment of Obstetrics and Fetal Medicine, University Medical Center Hamburg-Eppendorf, 20251 Hamburg, Germany

^bExperimental and Clinical Research Center (ECRC), a cooperation of the Max-Delbrück Center for Molecular Medicine (MDC) and Charité-Universitätsmedizin, 13125 Berlin, Germany

^cDepartment of Life Sciences, Imperial College London, London SW7 2AZ, UK

^dInstitute of Clinical Pharmacology and Toxicology, University Medical Center Hamburg-Eppendorf and German Centre for Cardiovascular Research (DZHK), Partner Site Hamburg/Kiel/Lübeck, 20249 Hamburg, Germany

^eClinical Neuroscience, Max Planck Institute for Multidisciplinary Sciences, 37075 Göttingen, Germany

^fDepartment of Cardiology and Nephrology, HELIOS-Klinikum, 13125 Berlin, Germany

^gDepartment of Pathology, Uniformed Services University of the Health Sciences, Bethesda, MD 20814, USA

*To whom correspondence should be addressed. Email: s.blois@uke.de

¹F.Z., N.F., and S.B. contributed equally to this work.

²Charité—Universitätsmedizin Berlin, corporate member of Freie Universität Berlin, Humboldt-Universität zu Berlin, and Berlin Institute of Health (BIH) and Institute of Biochemistry, Berlin, Germany and Deutsches Zentrum für Herz-Kreislauf-Forschung (DZHK), partner site Berlin, Germany. **Edited By:** Marisa Bartolomei

Abstract

Placental abnormalities cause impaired fetal growth and poor pregnancy outcome (e.g. preeclampsia [PE]) with long-lasting consequences for the mother and offspring. The molecular dialogue between the maternal niche and the developing placenta is critical for the function of this organ. Galectin-1 (gal-1), a highly expressed glycan-binding protein at the maternal–fetal interface, orchestrates the maternal adaptation to pregnancy and placenta development. Down-regulation or deficiency of gal-1 during pregnancy is associated with the development of PE; however, the maternal- and placental-derived gal-1 contributions to the disease onset are largely unknown. We demonstrate that lack of gal-1 imposes a risk for PE development in a niche-specific manner, and this is accompanied by a placental dysfunction highly influenced by the absence of maternal-derived gal-1. Notably, differential placental glycosylation through the Sda-capped N-glycans dominates the invasive trophoblast capacity triggered by maternal-derived gal-1. Our findings show that gal-1 derived from the maternal niche is essential for healthy placenta development and indicate that impairment of the gal-1 signaling pathway within the maternal niche could be a molecular cause for maternal cardiovascular maladaptation during pregnancy.

Keywords: galectin-1, maternal niche, preeclampsia, Sda antigen

Significance Statement

The placenta is formed from the molecular dialogue within the maternal niche; failure in placentation increases the risk of developing preeclampsia (PE), a life-threatening disease associated with lifelong adverse health consequences for the mother and offspring. Mechanisms underlying the maternal contribution to the placentation process remain unclear. This work uncovered the critical role of maternal galectin-1 (gal-1) in mammalian reproduction. Using specific niche Lgals1-deficient models, we determined that maternal-derived gal-1 boosts the invasive trophoblast capacity by modulating differential Sda terminal N-glycosylation and HB-EGF bioavailability at the maternal-fetal interface. These observations provide a better understanding of the maternal molecular dialogue involved in the placental adaptations required for a successful pregnancy.

Introduction

Preeclampsia (PE) is a complication of pregnancy defined by hypertension with or without proteinuria developed after the 20th week of gestation. This heterogeneous syndrome is a leading cause of maternal and neonatal mortality and morbidity worldwide (1). Due to the increased prenatal surveillance and early interventions, the mortality of PE has significantly dropped in several countries, including the United States. Nevertheless, the

Competing Interest: The authors declare no competing interest. **Received:** April 14, 2023. **Accepted:** July 20, 2023

OXFORD

© The Author(s) 2023. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

rising burden of PE short- and long-term cardiovascular sequelae impacts both maternal and child health (2, 3). The etiology of PE continues to be debated; however, research in this field has significantly contributed to understanding the disease pathogenesis, where the placenta component plays a central role (4, 5). Delivery of the placenta and the usually preterm baby remains the only available treatment; therefore, deciphering the underlying mechanisms of the disease is of great importance.

The placenta forms the maternal-fetal interface and determines fetal growth and survival supplying oxygen, nutrients, and waste exchange. Thus, impaired placenta function is generally associated with poor fetal outcomes and increased shedding of trophoblast microparticles into the maternal circulation (4). The inflammatory and antiangiogenic placenta shedding causes maternal endothelial cell activation, a hallmark of PE, and is responsible for the disease's clinical symptoms (5, 6). The role of the placenta as the trigger for the maternal syndrome has been extensively studied in the PE field (4, 7). However, the contribution of the maternal niche to the etiology of PE remains in question. Studies in humans profiling the chorionic villus have shown that impaired decidua maturation is associated with the development of PE (8). Moreover, human endometrial stromal cells isolated from women with severe PE failed to decidualize in vitro and did not support cytotrophoblast (CTB) invasion (9). These observations suggest that defects in the maternal niche could be a critical contributor to the placenta maladaptation triggering PE disease by diverse molecular pathways (10). However, molecular mechanisms governing the maternal-placental dialogue remain to be elucidated.

Previous studies by our and other groups showed that galectin-1 (gal-1), a member of the conserved family of soluble β-galactosidebinding proteins, is the most abundant galectin at the maternalfetal interface (11-14). Gal-1 is a key regulator of the placentalmaternal dialogue due to its ability to regulate critical processes, including maternal immune and vascular adaptation to pregnancy and trophoblast differentiation and invasion properties (13, 15-17). Indeed, gal-1-deficient pregnant dams developed a spontaneous PE-like syndrome characterized by hypertension, proteinuria, and fetal growth restriction (FGR) (17). Inhibition of endogenous gal-1 decreased the invasive capacity of human extravillous trophoblast affecting the spiral artery (SpA) remodeling process (16). Moreover, a decrease in circulating gal-1 levels between 8 and 10 weeks of gestation was observed in women who later developed PE (17). A reduced frequency of gal-1 expressing T and natural killer (NK) cells was reported in the peripheral blood of women diagnosed with PE, suggesting a role of this galectin in the development of PE (18). To gain further insight into the gal-1 maternal and placental contribution through which PE develops, we undertook an experimental in vivo model in which lack of gal-1 is conferred exclusively to the maternal or the placental niche (19). Here, we demonstrated that (i) gal-1 expression is higher in the maternal niche than the placental compartment during the period, (ii) gal-1 deficiency within the maternal niche sensitizes the cardiovascular adaptation to pregnancy and induces a PE-like syndrome, (iii) lack of maternal gal-1 signaling leads to impaired placenta function characterized by a distinguished glycopattern (e.g. loss of Sda-capped N-glycans) before PE onset, and (iv) maternal-derived gal-1 is a critical regulator of the heparinbinding epidermal growth factor-like growth factor (HB-EGF)/ β-1,4-N-acetylgalactosaminyltransferase 2 (B4GALNT2) loop responsible for the synthesis of Sda-capped glycans on invasive trophoblasts. We propose that the maternal niche, through gal-1, is a pivotal contributor to healthy placenta function and, therefore, to the etiology of PE. Therapies designed to improve

gal-1 expression in the maternal niche could be a novel strategy to prevent placenta pathologies with poor pregnancy outcomes.

Results

Maternal gal-1 deficiency sensitizes cardiovascular adaptation imposing a risk for PE development

We first analyzed the expression of gal-1 at the maternal and placental compartments during normal pregnancy. As shown in Fig. 1A, the levels of gal-1 are higher in the maternal niche compared with the placenta compartment at embryonic day (E)13 in wild-type (WT) dams. As experimental models to understand the importance of gal-1 in each compartment, we used in vitro fertilization to achieve either maternal gal-1 deficiency (mKO) or fetal-placental gal-1 deficiency (fplKO) (19). Lack of gal-1 expression within the decidua (mKO) and placenta (fplKO), respectively, are displayed in Fig. 1A. Previously, we showed that E13 fetuses carried by mKO dams suffered from FGR and fetal immaturity when compared with fplKO and WT dams (19). Here, we extended the analysis to the late gestation (E17) and postnatal (PN)28 period. We observed that placenta weights were lower and fetuses carried by fplKO and mKO dams displayed fetal immaturity (as shown by the lower Theiler stage distribution at E17) when compared with the WT dams (Fig. 1B and C). In addition, pups delivered by mKO dams showed significantly lower body weight when compared with fplKO mice on PN28 (Fig. 1D), suggesting that the altered fetal growth trajectory may affect the PN development. Alongside the observed FGR phenotype at E13, maternal systolic blood pressure was increased from E13 in mKO dams and was significantly higher compared with the fplKO and control dams at E17 (Fig. 1E). On the other hand, mKO displayed an increased urine albumin/creatinine ratio (ACR), but this was not statistically different when compared with fplKO and WT dams on E17 (Fig. 1F). Because hypertension is a clinical sign originating from endothelial dysfunction (1), we next characterized the cardiometric adaptation before disease onset. A normal endothelial function requires nitric oxide (NO) production from L-arginine (L-Arg). L-Arg can be methylated, and one of its derivatives, asymmetric dimethylarginine (ADMA), inhibits endothelial NO synthase (eNOS) activity (20). Elevated ADMA serum levels are associated with reduced vasodilation and endothelial dysfunction, and the ratio of L-Arg to ADMA is a valuable index to evaluate eNOS activity (21). Another L-Arg derivative, symmetric dimethylarginine (SDMA), also modifies endothelial function by competition with L-Arg transport without affecting eNOS activity (22). As shown in Fig. 1G, the L-Arg concentration is decreased, while the SDMA levels are increased in the maternal circulation of mKO compared with the WT. Furthermore, the ratio of L-Arg/ ADMA and L-Arg/SDMA is decreased in mKO, reflecting limited NO production. These findings suggest that gal-1 derived from the maternal niche is important for cardiovascular adaptation during pregnancy and impacts fetal maturity preceding the onset of maternal PE syndrome.

Lack of maternal-derived gal-1 has a significant impact on the placentation process

To further examine the effects of maternal-derived gal-1 at the maternal–fetal interface, we analyzed the post-placentation period (E13) prior to PE onset. The mature murine placenta consists of three layers: the labyrinth (Lab), the junctional zone (Jz), and the maternal decidua basalis (DB) (Fig. S1A). As shown in Fig. S1B,



Fig. 1. Deficiency of maternal gal-1 induces cardiovascular maladaptation in mice. A) Schematic depiction of gal-1 expression at the maternal–fetal interface and periphery in the experimental groups: $Lgals1^{+/+}$ (WT), fplKO and mKO (*left*). Gal-1 levels measured by ELISA in placenta and decidua tissue homogenates on embryonic day E13 (n = 6) (right). B) Placental weight (mg) was registered on E17. C) Fetal development defined by Theiler stage (TS) analysis was evaluated on fetuses derived from WT, fplKO, and mKO on E17. In mKO and fplKO pregnancies, higher proportion of fetuses were on TS 25 (P < 0.01 vs. WT, chi-square test), and lesser proportion reached the TS 26 (P < 0.01 vs. WT, chi-square test). D) Offspring body weight (grams) was registered on PN day 28. E) Systolic blood pressure (SBP) was measured from E8 to E17 (n = 5) by a noninvasive tail-cuff acquisition system (CODA System). F) Ratio of albumin to creatinine (ACR) in urine collected on E17 (n = 6). G) Measurement of L-Arg, SDMA, and the ratio of L-Arg/ADMA and L-Arg/SDMA in maternal circulation at E13 (n = 5) by MS/MS assay. In all figures, data are presented as the mean \pm SEM. In A) and G), *P < 0.05, **P < 0.01, and ***P < 0.001 using work ANOVA followed by Tukey's test; in E), *P < 0.05 using two-way ANOVA followed by Tukey's test, and in B) and D), *P < 0.05, **P < 0.01, and ***P < 0.001 using Kruskal–Wallis followed by Dunn's test.

we observed a reduction in the decidua area in mKO with less mature (DBA⁺PAS⁺) tissue-associated uterine NK cells (Fig. S1C). However, inflammation and areas of necrosis were apparent in the decidua area of both mKO and fplKO dams (Fig. S1D). Uterine vasculature transformation is a unique process that depends on uNK cells and invasive trophoblasts (4). Strikingly, SpA wall thickness (vessel/lumen ratio) (Fig. S1E), the number of mature vascular-associated uNK cells (Fig. S1E), and pancytokeratin⁺ trophoblast cells were reduced in both experimental models (Fig. S1E), suggesting that the SpA remodeling is influenced by the gal-1 derived from both niches. SpA remodeling is typically associated with the loss of pericytes (α -SMA positive cells) (6). We observed that the mKO decidua displayed an increased number of pericytes compared with WT and fplKO dams (Fig. S1F), indicating more severely impaired vascular remodeling in mice lacking gal-1 in the maternal compartment.

Further analysis revealed that the number of glycogen cells (GCs) in the Jz (PAS⁺ cells/mm²) is increased in fplKO, and mislocalized GC islets within the Lab layer of the placenta are evident in both fplKO and mKO mice (Fig. S2A). The Lab layer of the placenta is the site of nutrient exchange, and any defect may influence the ability to mediate the transfer of nutrients and oxygen to the fetus thus impacting fetal growth (23). Quantitative analysis of the fetal vasculature of the Lab (isolectin B4⁺) showed an increase in the total vessel area and branching, resulting in a reduced lacunarity in mKO dams (Fig. S2B). Interestingly, the fplKO Lab showed slightly reduced vasculature complexity (decreased area occupied by vessels, normal branching, and lacunarity). The placenta insufficiency phenotype at E13 is accompanied by an inflammatory trophoblast state (Fig. S2C). Thus, the augmented placenta inflammation in mKO dams suggests a decisive contribution of the maternal-derived gal-1 to the placentation process.

Since the feto-maternal interface simultaneously expresses several galectins, we posited that in the gal-1 knockout models, there could be functional compensation by other galectin family members. Therefore, to determine the galectin fingerprint in gal-1 knockout models, we used multiplex imaging to visualize the three most abundant galectins (gal-1, gal-3, and gal-9) in the post-placentation period (E13) (Fig. S2D). We observed that while in WT mice, uNK cells/decidual stromal cells expressed weak/ moderate levels of gal-3, respectively, the levels were higher in mKO and fplKO within the same cellular compartments. Analysis of gal-9 expression demonstrated a similar nuclear localization in all experimental groups (Fig. S2D). Concerning the trophoblasts that invaded the maternal decidua, mKO and fplKO showed a marked increase of gal-3 expression compared with the WT. Nuclear gal-9 expression was moderate in all groups and colocalized with gal-3. We conclude that mKO and fplKO compensated for the lack of gal-1 with gal-3 and gal-9 within the decidual compartment. However, distinct phenotypes in mKO and fplKO support the functional uniqueness role of gal-1 in the maternal niche.

Defective decidualization is known to influence the development of PE in humans (9); therefore, we next assessed the maternal vascular expansion on the pre-placentation period (E7) (Fig. S3A). Quantitative analysis of the vascular tree of the whole implantation site (endoglin⁺ vessels) showed a decrease in the total vessel area and of the total vessel length, resulting in a reduced vascular expansion in mKO dams when compared with WT dams (Fig. S3B). Comparison of the decidualization translational genes revealed a significant decrease of Prl8a2 and Wnt5a decidual expression in mKO dams compared with WT (Fig. S3C). This was accompanied by an aberrant uNK activation (degranulated perforin⁺ NK cells) in the pre-placentation period (Fig. S3D). These results suggest that lack of maternal gal-1 during decidualization caused inadequate vascular expansion, differential NK activation, and decrease in decidualization makers which may influence trophoblast invasion into the decidua.

Loss of placental Sda-capped glycans anticipates the development of PE-like syndrome

Placenta insufficiency can result from delayed trophoblast differentiation (24). To determine whether the lack of maternal-derived gal-1 could impact the differentiation stage of the trophoblast subsets, we assessed the gene expression profile that characterized the Jz and Lab layers of the placenta at E13 before PE onset. mKO-derived placentas showed a reduction of *Hand-1*, an essential transcription factor for trophoblast giant cell differentiation, and of placental lactogen I (*Prl3d1*), which is expressed by terminally differentiated trophoblast giant cells within the Jz compared with WT and fplKO-derived placentas (Fig. 2A). This indicates that impaired maternal-derived gal-1 signaling delayed placenta differentiation of the Jz. However, we observed similar levels of expression of genes involved in Lab development.

Glycans modify proteins required for trophoblast function, and previous studies showed that altered glycosylation is associated with placental insufficiency (25-27). To determine whether the placenta glycophenotype differed in the mKO dams at E13, we used a MALDI-MS/MS-based glycomic approach (28, 29) to define the N-glycan profiles of placental tissue from WT, fplKO, mKO, and fullKO mice. All samples contained similar N-glycome profiles in the low to mid mass range (m/z 1,500-3,500), where high mannose and biantennary complex-type N-glycans are abundant (Figs. 2B and S4). However, clear differences were observed in the high mass range (m/z 3,500-6,000) in the absence of gal-1, as illustrated in Fig. 2B. Comparable glycomic profiles were found in three biological replicates (Figs. S4 and S5). The WT, fplKO and mKO spectra are characterized by a molecular ion series at m/z 3,947, 4,600, 5,002, 5,411, and 5,812. MS/MS experiments revealed that these components are complex, core-fucosylated tri- and tetra-antennary glycans with up to four antenna capped by the Sda epitope GalNAc β 1–4(NeuAc α 2-3)Gal β 1-4GlcNAc (Figs. 2B, S4, and S5). Additional antennae are either α -Gal capped or are truncated at GlcNAc (see cartoon annotations in Fig. 2C). Thus m/z 3,947 and m/z 4,600 have two Sda antennae, m/z 5,002 and 5,411 have three Sda antennae, and m/z 5,812 has four Sda antennae. A single α -Gal-capped antenna is found in the m/z 4,600 and m/z 5,812 glycans. The overall spectra for WT, fplKO, and mKO in this mass range closely resemble each other, although that of the mKO has somewhat reduced levels of the Sda-containing glycans compared with WT, while the Sda abundance in fplKO is higher (Figs. 2B, S4, and S5). In contrast, Sda-containing glycans are either absent or low in abundance in the fullKO. Instead, its glycome is dominated by NeuGc-capped glycans in the high mass range (Fig. 2B). Therefore, we conclude that gal-1 contributes significantly to Sda regulation since the absence of this lectin in the fullKO is associated with an almost complete loss of Sda epitopes. These results constitute additional evidence of the maternal-derived gal-1 contribution to placental differential glycosylation which antedates the development of PE.

The last step in the biosynthesis of Sda-capped glycans is catalyzed by B4GALNT2, and inhibition of this enzyme impaired mouse embryo implantation (30). As shown in Fig. 2D, expression of B4GALNT2 is down-regulated in the decidua and trophoblast GC layer of gal-1 fullKO placenta, suggesting that gal-1 may regulate B4GALNT2 expression on the invasive trophoblast lineages. Knowing that endogenous gal-1 also is involved in the invasive machinery of trophoblast (13, 15–17), we investigated a recently published single-cell RNA-sequencing (RNA-seq) analysis of placenta and decidua biopsies to explore a possible association of LGALS1 expression levels with trophoblast differentiation trajectory in humans (31). To this end, we assessed the cells annotated as "trophoblast" from the original study and constructed their pseudotime trajectory using the "Monocle2" tool. Consistent with previous results (15, 17), extravillous CTBs (EVT) bifurcate from the trajectory of CTB to syncytiotrophoblasts (STB) and are placed on a single branch toward the end of the pseudotime trajectory of trophoblast (see Materials and Methods S1). This recapitulated the biological trajectory of trophoblast differentiation. Upon examining the LGALS1 transcript expression, we observed that was abundantly and specifically expressed in the EVT lineage, while showing higher expression along the trajectory of EVT (Fig. 2E). Next, we asked if the trophoblast of women suffering from PE have differential expression in comparison with healthy counterparts. Indeed, EVT derived from PE women showed a significant down-regulation of LGALS1 when compared with women that have a normal pregnancy (Fig. 2E). Overall, these results are consistent with our hypothesis that decidual gal-1 deregulation is associated with PE disease.

Gal-1 facilitates trophoblast invasive properties through Sda-capped glycans and HB-EGF maturation

To further characterize the role of Sda-capped glycans in pregnancy, we inhibited the B4GALNT2 enzyme by siRNA in the murine SM9–1 and human HTR-8/SVneo trophoblast cell lines (Figs. 3A and S6A and B, Table S1 and S2). Reduction of B4GALNT2 levels in the mouse SM9-1 and human HTR-8/SVneo trophoblast cell lines significantly impaired their invasion capacity (Fig. 3A and S6C). In addition, treatment of SM9-1 cells with exogenous gal-1 increased the expression of B4GALNT2 (Fig. 3B). These results indicated that gal-1 increased the activity of B4GALNT2 and that this enzyme is involved in the invasion



Fig. 2. Absence of placental Sda-capped glycans is associated to gal-1 deficiency. A) Heatmap showing the expression of trophoblast developmental genes specific for the Jz and Lab layers, analyzed by qPCR followed by Z-score normalization (n = 7), [#]P < 0.05 mKO vs. WT and fplKO; ^{*}P < 0.05 mKO vs. WT using Kruskal–Wallis followed by Dunn's test. B) MALDI-TOF MS spectra of N-glycans in the high mass range (m/z 3,500–6,000) extracted from bulk placental tissue derived from WT, fplKO, mKO, and fulKO dams on embryonic day E13. C) MALDI-TOF/TOF MS/MS fragmentation analysis of [M + Na] + peak m/z 5,812, illustrating the loss of Sda antigen. Man, mannose; Gal, galactose; GalNAc, N-acetylgalactosamine; GlcNAc, N-acetylglucosamine; NeuAc, N-acetylneuraminic acid; Fuc, Fucose. D) Representative images of B4GALNT2 immunohistochemistry staining within the decidua compartment (upper) and giant cell layer (lower) of the placenta on E13. Scale bar: 20 µm. In the decidua, uterine NK cells ([#]), invasive trophoblast (asterisk), and endthelial cell (arrow) are shown. The giant cell layer (delineated by dashed lines) is located between the decidua (Dec) and Jz. Quantification of the expression of B4GALNT2 on invasive trophoblast (upper right) or parietal trophoblast giant cells (TGC, lower right) expressed as H-score, ^{***}P < 0.001 using two-tailed t test. E) Pseudotime trajectory showing the ordering of major trophoblast lineages at the maternal–fetal interface (upper (left)) was performed from scRNA-seq data set analysis. Monocle2 visualization of three distinct clusters of trophoblasts using the top 2,000 differentially expressed genes projected into a 2D space. Codes represent CTB (middle), syncytiotrophoblast (STB, left), and extravillous trophoblast (EVT, right). Top right panel illustrates intensity and abundance of LGALS1 transcript expression along the trajectory in previous figure. Colors represent an average Log2 expression level scaled to the number of unique molecular identification (UMI) values in single c



Fig. 3. Exogenous gal-1 is involved in trophoblast invasion through Sda-capped glycans and sHB-EGF. A) Schematic drawing of murine trophoblast cell line SM9-1 culture condition with B4galnt2 siRNA. The invasion capacity of SM9-1 cells was evaluated by transwell invasion assay (left). Percentage of invaded SM9-1 cells (middle) and representative images (right) of SM9-1 with siRNA for B4galnt2 (siRNA) or a scramble sequence (Scr) as control (n = 3, 5 pictures/assay), **P < 0.01 using two-tailed t test. B) Experimental designs of the murine trophoblast cell line SM9-1 treatment with recombinant gal-1 (rGal-1) after 24-h B4GALNT2 expression and sHB-EGF levels in the culture supernatant were analyzed by ELISA. B4GALNT2 protein expressions were tested by western blot in SM9-1 cells upon stimulation with 0, 10, or 100 ng/mL of rgal-1 in three independent experiments. \pounds -ACTIN was used to calculate relative B4GALNT2 protein expression levels. C) Circulating progesterone levels measured on E13 in WT, fplKO, and mKO dams by ELISA (n = 7). D) sHB-EGF ELISA determination in the supermatant derived from SM9-1 cells stimulated with 0, 10, and 100 ng/mL of rgal-1 for 24 h (n = 6). E) Primary trophoblasts derived from Lgals1^{+/+} (WT) or Lgals1^{-/-} (fullKO) placentas on E13 were isolated by Percoll gradient, and B4GALNT2 expression was evaluated upon treatment with recombinant HB-EGF (0, 0, 1, 1, or 10 nM HB-EGF) for 24 h using IHC. H-score was obtained from the analysis of three independent experiments (five images/experimental condition). In all figures, data are plotted as the mean \pm SEM. In B)–E), *P < 0.05, **P < 0.01, ***P < 0.001, and ****P < 0.001 using Kruskal-Wallis followed by Dunn's test.

capacity of trophoblast cells, a critical process necessary for the maternal vasculature transformation. To identify if progesterone, a known inducer of B4GALNT2 expression (30), was downregulated in the gal-1-deficient pregnancies, we determined the maternal levels at E13. This analysis revealed that gal-1 deficiency in the maternal niche is associated with decreased progesterone in maternal circulation (Fig. 3C). Outside reproductive tissues, soluble heparin-binding epidermal growth factor-like growth factor (sHB-EGF) induces the expression of B4GALNT2 (32). Of note, HB-EGF is expressed at high levels and was associated with trophoblast cells' survival and invasive capacity during the first trimester of pregnancy (33). Interestingly, placentas derived from PE patients displayed a decreased level of HB-EGF (34). Based on our observation that gal-1 is responsible for the production of mature HB-EGF, we hypothesized that exogenous gal-1 is involved in activating HB-EGF on trophoblast cells. To test this hypothesis, we evaluated the level of sHB-EGF in trophoblasts upon gal-1 stimulation in vitro. As shown in Fig. 3D, exogenous gal-1 induced the maturation of HB-EGF in a dose-dependent manner. Interestingly, the level of sHB-EGF is significantly decreased in unstimulated trophoblast cells from fullKO mice compared with WT (Fig. S6D). Of note, stimulation of *Lgals1*-deficient trophoblasts with sHB-EGF resulted in similar B4GALNT2 expression levels than in *Lgals1* WT trophoblasts (Fig. 3E). Thus, exogenous gal-1 is necessary for the maturation of HB-EGF at the maternal-fetal interface; however, expression of B4GALNT2 can also be induced by sHB-EGF independently of the intracellular gal-1 expression.

Discussion

For several decades, the contribution of the maternal niche to the etiology of PE has been unappreciated; however, recent studies utilizing in vitro approaches and global transcriptional profiling highlighted its critical importance. Nevertheless, the mechanisms by which the maternal niche impacts the placentation program are poorly understood. Galectins, particularly gal-1, contribute to maternal and placental niches' healthy adaptation to pregnancy (11, 12). Interestingly, early studies noted that the lack of gal-1 in both compartments imposed a risk for PE development (17). In this study, we have used in vivo models to determine how maternal- and placental-derived gal-1 deficiency impacts pregnancy outcomes. We noted that gal-1 is more abundant in

the maternal than the placental niche once pregnancy is established. The lack of gal-1 in the maternal compartment severely interferes with placenta efficiency, which results in an altered fetal maturity trajectory accompanied by a PE-like syndrome. In addition, our findings show that maternal-derived gal-1 regulates the invasive trophoblast capacity through the maturation of HB-EGF, which increases Sda terminal glycosylation (Fig. 4). Compensation by other galectin members (gal-3 and gal-9) is insufficient to overcome gal-1 absence in the maternal niche. Thus, we provide evidence of the unique function of the maternal compartment through this lectin in orchestrating a healthy pregnancy.

Particular attention has been paid over the last decades to the role of gal-1 at the maternal-fetal interface. In vitro studies carried out to understand the function of the placental-derived gal-1 (endogenous) showed that this lectin is required for the trophoblast invasive and immunoregulatory properties (e.g. HLA-G expression) necessary for optimal placentation (16). Furthermore, we previously showed a dramatic placenta insufficiency leading to FGR and PE development in fullKO (maternal/ placental) dams (17). The results presented here using in vivo models demonstrate that loss of placental gal-1 (fplKO) has a modest impact on fetal maturity and placenta growth with no maternal manifestations of PE-like syndrome. Thus, inadequate placental-derived gal-1 is not sufficient to pose a threat to the pregnancy. On the other hand, this study revealed that gal-1 WT placentas were not fully functional when gal-1 was absent in the maternal niche, as mKO dams spontaneously developed lategestational inadequate cardiovascular adaptation (clinical features of PE). Our findings provide evidence that maternal-derived gal-1 (exogenous) signaling is necessary to maintain a functional Lgals1-deficient placenta. Interestingly, we previously showed that dams in which gal-1 is blocked with Anginex, a specific gal-1 antagonist, are at risk of developing PE, mainly due to a dramatically abnormal trophoblast vascular remodeling profile (17). Indeed, our analysis of the placenta developmental stage on mKO dams showed a significant decrease in Hand-1 and Prl3d1 gene expression, which mainly characterize the Jz. This heterogeneous placenta layer plays a pivotal role in decidual invasion and SpA remodeling, among others. Notably, Hand-1 mutant trophoblast cells are less invasive in vitro (35, 36), and Prl3d1 is involved in maternal metabolism and maintains a constant supply of glucose and amino acids to the fetus (37). In line with these observations, mKO-derived placentas showed delayed trophoblast maturity involved in the transformation of uterine vasculature (SpA remodeling) which is associated with the development of PE (38). Uterine NK cells and invasive trophoblasts are critical for maternal vasculature remodeling during pregnancy (4). Therefore, we analyzed the impact of maternal-derived gal-1 deficiency on the tissue- and vascular-associated uNK cells. Decreased uNK cells associated with SpA walls in mKO dams agree with the exacerbated placental hypoxia. This is further amplified by the severe inflammatory placental phenotype in these mice, which manifests as inadequate maternal cardiovascular adaptation. Indeed, PE is associated with exacerbated immune activation caused by the imbalance of proinflammatory/ regulatory/antiinflammatory cytokines which further promotes the endothelial dysfunction and subsequent clinical disease manifestation (39). These results provide further proof that a deficit of maternal-derived gal-1 is sufficient to cause PE disease.

During pregnancy, physiologic changes occur in the mother's circulation to meet the needs of the developing fetus. Early in pregnancy, an increase in the maternal blood volume and cardiac

output without increasing the blood pressure is observed. This phenomenon is due to a general decrease in vascular resistance and is attributed to the remodeling of the spiral arteries (40). Remarkably when the vessel architecture of the uterus is developed, the increase in blood flow is compensated by sustained vasodilation of these existing vessels (41). Therefore, from mid-gestation, adequate uteroplacental blood flow depends on vasodilation and early-onset PE can be characterized by reduced capacity for vasodilation (1, 42). One of the principal vasodilators is NO, and its production is catalyzed by eNOS from L-Arg in the maternal endothelium (43). Moreover, NO is essential for the invasion of the CTBs into the maternal SpA and a reduction of NO biosynthesis or bioavailability has been associated with PE (44). Our results demonstrated a significant decrease in L-Arg and an increase in SDMA, an inhibitor of L-Arg transport (22) in mKO. The ratio L-Arg/ADMA, an index to measure eNOS activity (21), and the ratio L-Arg/SDMA are also down-regulated in mKO. Overall, these results showed a decreased production of NO in mKO. In these animals, placenta inflammation and an increased blood pressure were observed later in gestation, suggesting an impaired vascular adaptation due to the lack of maternal gal-1 is associated with PE-like syndrome. Like our results, a decrease of L-Arg levels and in the ratio L-Arg/ADMA was observed in patients with PE compared with controls (45, 46). We have previously demonstrated that in early gestation, the inhibition of gal-1-mediated angiogenesis by Anginex provokes PE-like symptoms (17). Normal pregnancy is associated with endothelium-dependent vascular adaptation, and especially gal-1 is a critical component of the vascular remodeling (47, 48). Our study design allowed us to uncover the role of gal-1 within the maternal vascular adaptations to pregnancy and its role in the Lab layer phenotype. Consistent with the maternal vascular remodeling defects in mKO mice, the fetal Lab showed increased branching morphogenesis and reduced lacunarity. It is important to note that Lab layer changes in the vascular compartment coincided with lower placental partial pressure of oxygen (PO2) in Lgals1-deficient mice (49), which directly links poor placental oxygenation and placental dysfunction and is consistent with the previously reported gal-1 multiple endothelial cell functions, including activation, proliferation, migration, tube formation, and sprouting at the maternal-fetal interface (50). Our results further highlight the importance of maternalderived gal-1 in the development of PE, since the lack of gal-1 in this compartment, and not in the placenta, showed a severely impaired endothelial function.

The maternal niche, through several factors, tightly controls the invasive trophoblast capacity to enable proper placentation (51). Failure in trophoblast invasion and reduced SpA remodeling is associated with pregnancies complicated with PE (52). We have previously shown differential glycosylation of human trophoblast cell types, including the enrichment of polylactosaminecontaining N-glycans that could act as receptors for galectins, on EVT involved in SpA remodeling (53). Notably, our present study revealed that mKO-derived placenta displayed altered glycosylation characterized by a reduction of complex N-glycans capped by the Sda epitope. The enzyme involved in Sda synthesis is B4GALNT2, previously known as β1,4GalNAcT-II (GALGT2) (54). This enzyme is up-regulated by progesterone in the mouse uterus and, during pregnancy, peaks at E10, coinciding with trophoblast invasion into the decidua (30, 55). Our findings of reduced levels of progesterone in fplKO and mKO dams compared with controls, combined with evidence of B4GALNT2 regulation by this hormone, highlight the significance of the cooperation between



Fig. 4. Schematic depiction of the maternal gal-1 effects on the regulation of trophoblast invasion and remodeling of the SpA. In normal pregnancy, maternal gal-1 induces sHB-EGF secretion, leading to B4GALNT2 activation and Sda-capped glycosylation. As a result, trophoblasts acquire an invasive phenotype necessary for SpA remodeling. In preeclampsia, reduced maternal gal-1 impairs sHB-EGF maturation by directly affecting B4GALNT2 activation and indirectly by altering the concentration of progesterone, a hormone that increases the availability of sHB-EGF. Inadequate trophoblast invasive capacity compromises the SpA remodeling and subsequent preeclampsia development.

progesterone-gal-1 regulation pathways during placenta development; this synergy is also relevant in the immune adaptations required for a successful pregnancy (14, 15). However, whether the differential glycosylation (Sda-capped glycans) is involved in the invasive trophoblast capacity has not been determined. Our data demonstrate that the inhibition of B4GALNT2 reduced the invasion capacity of the murine SM9-1 and human HTR-8/SVneo trophoblast cell lines, suggesting a role of Sda terminal glycosylation on the invasive properties of trophoblast cells at the maternal-fetal interface. In the present study, we also showed that EVT LGALS1 expression is dysregulated during PE as indicated by single-cell RNA-seq analysis, which, for the first time, associates the decidual gal-1 with the PE syndrome. We have previously suggested additional functions for the Sda epitope on (i) the murine zona pellucida 3 glycoprotein, which participates in sperm-egg binding (cell–cell interaction) (56), and (ii) bovine pregnancy-associated glycoproteins where the differential expression of Sda during pregnancy may influence the serum halflife of these proteins (57). Thus, the synthesis of the Sda-glycotope is likely regulated in a tissue-specific manner and could be under hormonal regulation. In this regard, it has been shown that HB-EGF regulates B4GALNT2 expression in mouse skeletal muscle (32). HB-EGF participates in pregnancy-associated processes, including embryo-uterine cross-talk during implantation in mice and humans (58, 59). Decreased placental HB-EGF was observed in pregnancies complicated with PE (34, 60). Here, we show that exogenous sHB-EGF induced the expression of B4GALNT2 in vitro of primary isolated trophoblast cells, indicating that HB-EGF activates the synthesis of the Sda-glycotope at the maternal-fetal interface. Interestingly, gal-1 induces HB-EGF expression and ectodomain cleavage increasing its bioavailability in the lung tumor microenvironment (61). We observed that exogenously added gal-1 induced the maturation of HB-EGF in a murine trophoblast cell line and B4GALNT2 expression. Thus, our results demonstrated that maternal-derived gal-1 activates the trophoblast proinvasive program through HB-EGF maturation and differential Sda terminal glycosylation. This molecular

pathway may be involved in one of the several identified gal-1 protective functions during pregnancy.

In summary, we demonstrated that the maternal niche through gal-1 expression modulates the placental functional development and that this lectin promotes maternal cardiovascular adaptation during pregnancy. Furthermore, our findings suggest that designing therapeutic strategies to improve maternal gal-1 expression could improve pregnancy outcomes and reduce the increased cardiovascular risk post-preeclampsia.

Materials and methods Mice and tissue collection

Lgals1^{+/+} and Lgals1^{-/-} mice (129/P3J background) were acquired and maintained, and in vitro fertilization timed pregnancies were established as described (19). Lgals1^{+/+} embryos were transferred to $Lgals1^{-/-}$ dams to create the mKO, whereas a fplKO was obtained by transferring Lgals1^{-/-} embryos to Lgals1^{+/+} recipient dams. Animal care and all experimental procedures were performed according to Charité/University Medical Center Hamburg-Eppendorf institutional guidelines and conform to requirements of the German Animal Welfare Act (Berlin/ Hamburg). Timed pregnant mice were evaluated at embryonic day (E) 7, 13, and E17 (n = 6-8 mice per group). Offsprings from the different experimental models were evaluated on PN day 28. Whole implantations were frozen for cryostat sections or formalin fixed for paraffin sections on E7 and E13. Placental and decidual tissues were separated on E13 and frozen for isolation of total protein or total RNA according to standard procedures (15). Placental weight was recorded in late gestation (E17). Fetal development was analyzed by Theiler stage and body weight as described previously (62). Details are available in Materials and Methods S1.

ELISAs

Quantification of murine gal-1 and sHB-EGF were performed using the mouse gal-1 DuoSet enzyme-linked immunosorbent assay

(ELISA) (R&D Systems; DY1245) and mouse HB-EGF DuoSet ELISA (R&D Systems; DY8239) respectively, following the manufacturer's recommendations. Details are available in Materials and Methods S1.

Cardiovascular adaptation analysis in pregnant mice

Systolic and diastolic pressure was measured in the tail artery in pregnant females from E8 to E17 with a computerized, noninvasive tail-cuff acquisition system (CODA System; Kent Scientific) as described previously (16). In addition, L-Arg, ADMA, and SDMA were quantified from blood samples stored at -80°C applying a validated tandem mass spectrometric assay (MS/MS) (63). Details are available in Materials and Methods S1.

N-Glycan profiling of placental samples

N-glycan analysis from mouse placenta was carried out by MALDI-TOF/TOF according to the previous study (64). Details are available in Materials and Methods S1.

Cell culture and in vitro treatments

B4GALNT2 expression was inhibited in the mouse SM9-1 and human HTR-8/SVneo trophoblast cell lines by siRNA (Origene, SR416377 and SR314851, respectively) according to the manufacturer's instructions. After 36 h, cells were harvested to analyze B4GALNT2 mRNA or protein expression or to perform the invasion assay. For gal-1 treatments, 6-h FBS starvation was previously performed in SM9-1 cells and then incubated with 10 or 100 ng/ ml recombinant gal-1 (Peprotech, 450–39). After 24 h, supernatant was collected, and cells were harvested to protein isolation and western blot analysis. Trophoblast cells were isolated from E13 placenta as previously described (65). A 6-h serum starvation was performed, and then trophoblasts were treated with 0.1, 1, or 10 nM of recombinant HB-EGF (R&D, 259-HE/CF). After 24 h, cells were collected and cytospin was performed to evaluate B4GALNT2 expression. Details are available in Materials and Methods S1.

Statistics

Numerical data were analyzed with D'Agostino–Pearson normality test followed by ANOVA and Tukey's test or Kruskal–Wallis analysis and Dunn's test. Categorical variables are presented as frequency and assessed by chi-square test with post hoc Bonferroni adjustment. Statistical significance was designated as P < 0.05 analyzed by GraphPad Prism 9.0.

Acknowledgments

We thank T. Andreas, E. Kittmann (Blois's Lab at UKE), P. Moschansky, M. Daniltchenko, Y. Almousa, G. Barrientos (Blois's lab at Charité-Campus Virchow Klinikum), G. Koch (Blois's lab at Charité-Campus Buch), I. Urban, G. Michel (Charité-Transgene Facility), and U. Matschl (Heinrich-Pette-Institute) for their excellent technical assistance in generating this work. The content is solely the responsibility of the authors and does not necessarily represent the official views of the NIH or the US Department of Defense. Figures 1A; 3A, B, and E; and 4; S1A; and S3A were created with BioRender.com.

Supplementary material

Supplementary material is available at PNAS Nexus online.

Funding

This work was supported by grants from the Deutsche Forschungsgemeinschaft (DFG) BL1115/4-1, Heisenberg Program (BL1115/3-1, BL1115/7-1, and BL1115/11-1), and Heike Wolfgang Mühlbauer Stiftung to S.M.B; DFG DE631/15-1 to R.D.; Biotechnology and Biological Sciences Research Council (grant number BB/K016164/1) to A.D. and S.M.H.; and National Institute of Allergy and Infectious Diseases–NIH Grant R21AI156058 to G.D. Else Kröner-Fresenius-Stiftung (Project 2017_A123) to N.F., Y.X., F.Z., and Y.W. was supported by China Scholarship Council (CSC) Program.

Author contributions

S.M.B. conceived the research; S.M.B., M.G.G., A.D., S.M.H., R.D., and G.D. designed the research; Y.X., F.Z., N.F., S.B., Y.W, C.H., P.-C.P., J. D., T.W., E.S., M.S., M.G.G, and S.M.B. performed the research; Y.X., F.Z., N.F., S.B., Y.W, C.H., P.-C.P., J. D., T.W., E.S., M.S., A.D., S.M.H., G.D., M.G.G, and S.M.B. analyzed the data; and Y.X., M.G.G., and S.M.B. wrote the paper.

Data availability

All the codes to perform and reproduce the single-cell analysis presented in this manuscript are available at https://github.com/Manu-1512/Galectin-in-placenta-development-and-preecla mpsia.

References

- 1 Chappell LC, Cluver CA, Kingdom J, Tong S. 2021. Pre-eclampsia. Lancet 398:341–354.
- 2 Gammill HS, et al. 2018. Cardiomyopathy and preeclampsia. Circulation 138:2359–2366.
- 3 Shih T, et al. 2016. The rising burden of preeclampsia in the United States impacts both maternal and child health. Am J Perinatol. 33:329–338.
- 4 Fisher SJ. 2015. Why is placentation abnormal in preeclampsia? Am J Obstet Gynecol. 213:S115–S122.
- 5 Rana S, Lemoine E, Granger JP, Karumanchi SA. 2019. Preeclampsia: pathophysiology, challenges, and perspectives. *Circ Res.* 124:1094–1112.
- 6 Burton GJ, Redman CW, Roberts JM, Moffett A. 2019. Pre-eclampsia: pathophysiology and clinical implications. BMJ. 366:12381.
- 7 Zadora J, et al. 2017. Disturbed placental imprinting in preeclampsia leads to altered expression of DLX5, a human-specific early trophoblast marker. Circulation 136:1824–1839.
- 8 Rabaglino MB, Post Uiterweer ED, Jeyabalan A, Hogge WA, Conrad KP. 2015. Bioinformatics approach reveals evidence for impaired endometrial maturation before and during early pregnancy in women who developed preeclampsia. *Hypertension* 65: 421–429.
- 9 Garrido-Gomez T, et al. 2017. Defective decidualization during and after severe preeclampsia reveals a possible maternal contribution to the etiology. Proc Natl Acad Sci U S A. 114: E8468–E8477.
- 10 Than NG, et al. 2022. Early pathways, biomarkers, and four distinct molecular subclasses of preeclampsia: the intersection of clinical, pathological, and high-dimensional biology studies. *Placenta* 125:10–19.

- 11 Blois SM, et al. 2019. Pregnancy galectinology: insights into a complex network of glycan binding proteins. Front Immunol. 10:1166.
- 12 Menkhorst E, et al. 2021. Medawar's PostEra: galectins emerged as key players during fetal-maternal glycoimmune adaptation. Front Immunol. 12:784473.
- 13 Kopcow HD, et al. 2008. T cell apoptosis at the maternal-fetal interface in early human pregnancy, involvement of galectin-1. Proc Natl Acad Sci U S A. 105:18472–18477.
- 14 Than NG, et al. 2008. Emergence of hormonal and redox regulation of galectin-1 in placental mammals: implication in maternal-fetal immune tolerance. Proc Natl Acad Sci U S A. 105: 15819–15824.
- 15 Blois SM, et al. 2007. A pivotal role for galectin-1 in fetomaternal tolerance. Nat Med. 13:1450–1457.
- 16 Tirado-Gonzalez I, et al. 2013. Galectin-1 influences trophoblast immune evasion and emerges as a predictive factor for the outcome of pregnancy. Mol Hum Reprod. 19:43–53.
- 17 Freitag N, et al. 2013. Interfering with Gal-1-mediated angiogenesis contributes to the pathogenesis of preeclampsia. Proc Natl Acad Sci U S A. 110:11451–11456.
- 18 Molvarec A, et al. 2011. Peripheral blood galectin-1-expressing T and natural killer cells in normal pregnancy and preeclampsia. Clin Immunol. 139:48–56.
- 19 Borowski S, et al. 2022. Examination of the contributions of maternal/placental-derived galectin-1 to pregnancy outcome. Methods Mol Biol. 2442:603–619.
- 20 Böger RH. 2006. Asymmetric dimethylarginine (ADMA): a novel risk marker in cardiovascular medicine and beyond. Ann Med. 38:126–136.
- 21 Bode-Böger SM, Scalera F, Ignarro LJ. 2007. The L-arginine paradox: importance of the L-arginine/asymmetrical dimethylarginine ratio. Pharmacol Ther. 114:295–306.
- 22 Braekke K, Ueland PM, Harsem NK, Staff AC. 2009. Asymmetric dimethylarginine in the maternal and fetal circulation in preeclampsia. *Pediatr Res.* 66:411–415.
- 23 Woods L, Perez-Garcia V, Hemberger M. 2018. Regulation of placental development and its impact on fetal growth-new insights from mouse models. Front Endocrinol (Lausanne). 9:570.
- 24 Longtine MS, Nelson DM. 2011. Placental dysfunction and fetal programming: the importance of placental size, shape, histopathology, and molecular composition. *Semin Reprod Med.* 29: 187–196.
- 25 Sukhikh GT, et al. 2016. Differences of glycocalyx composition in the structural elements of placenta in preeclampsia. Placenta 43: 69–76.
- 26 Blois SM, Prince PD, Borowski S, Galleano M, Barrientos G. 2021. Placental glycoredox dysregulation associated with disease progression in an animal model of superimposed preeclampsia. *Cells* 10:800.
- 27 Tannetta D, Masliukaite I, Vatish M, Redman C, Sargent I. 2017. Update of syncytiotrophoblast derived extracellular vesicles in normal pregnancy and preeclampsia. J Reprod Immunol. 119: 98–106.
- 28 Hautala LC, et al. 2020. Altered glycosylation of glycodelin in endometrial carcinoma. Lab Invest. 100:1014–1025.
- 29 Mendoza M, et al. 2020. Glycan characterization of pregnancyspecific glycoprotein 1 and its identification as a novel Galectin-1 ligand. Glycobiology 30:895–909.
- 30 Li PT, Liao CJ, Wu WG, Yu LC, Chu ST. 2011. Progesterone-regulated B4galnt2 expression is a requirement for embryo implantation in mice. *Fertil Steril.* 95:2404–2409, 2409 e2401-2403.

- 31 Vento-Tormo R, et al. 2018. Single-cell reconstruction of the early maternal-fetal interface in humans. Nature 563:347–353.
- 32 Cramer ML, Xu R, Martin PT. 2019. Soluble heparin binding epidermal growth factor-like growth factor is a regulator of GALGT2 expression and GALGT2-dependent muscle and neuromuscular phenotypes. *Mol Cell Biol.* 39:e00140-19.
- 33 Leach RE, et al. 2004. Heparin-binding EGF-like growth factor regulates human extravillous cytotrophoblast development during conversion to the invasive phenotype. Dev Biol. 266:223–237.
- 34 Armant DR, et al. 2015. Reduced expression of the epidermal growth factor signaling system in preeclampsia. Placenta 36: 270–278.
- 35 Riley P, Anson-Cartwright L, Cross JC. 1998. The Hand1 bHLH transcription factor is essential for placentation and cardiac morphogenesis. Nat Genet. 18:271–275.
- 36 Hemberger M, Hughes M, Cross JC. 2004. Trophoblast stem cells differentiate in vitro into invasive trophoblast giant cells. *Dev* Biol. 271:362–371.
- 37 Berga SL, Nitsche JF, Braunstein GD. Endocrine changes in pregnancy. 2016. In: Melmed S, Polonsky K, Larsen PR, Kronenberg H, editors. Williams textbook of endocrinology. Elsevier. p. 831–848.
- 38 Dokras A, et al. 2006. Severe feto-placental abnormalities precede the onset of hypertension and proteinuria in a mouse model of preeclampsia. Biol Reprod. 75:899–907.
- 39 Harmon AC, et al. 2016. The role of inflammation in the pathology of preeclampsia. Clin Sci (Lond). 130:409–419.
- 40 Ramlakhan KP, Johnson MR, Roos-Hesselink JW. 2020. Pregnancy and cardiovascular disease. Nat Rev Cardiol. 17:718–731.
- 41 Boeldt DS, Bird IM. 2017. Vascular adaptation in pregnancy and endothelial dysfunction in preeclampsia. J Endocrinol. 232: R27–R44.
- 42 Krupp J, et al. 2013. The loss of sustained Ca(2+) signaling underlies suppressed endothelial nitric oxide production in preeclamptic pregnancies: implications for new therapy. Am J Physiol Heart Circ Physiol. 305:H969–H979.
- 43 McElwain CJ, Tuboly E, McCarthy FP, McCarthy CM. 2020. Mechanisms of endothelial dysfunction in pre-eclampsia and gestational diabetes mellitus: windows into future cardiometabolic health? Front Endocrinol (Lausanne). 11:655.
- 44 Khalil A, Hardman L, O'Brien P. 2015. The role of arginine, homoarginine and nitric oxide in pregnancy. Amino Acids 47: 1715–1727.
- 45 Kim YJ, et al. 2006. Reduced L-arginine level and decreased placental eNOS activity in preeclampsia. *Placenta* 27:438–444.
- 46 Speer PD, et al. 2008. Elevated asymmetric dimethylarginine concentrations precede clinical preeclampsia, but not pregnancies with small-for-gestational-age infants. Am J Obstet Gynecol. 198: 112.e1–112.e7.
- 47 Thijssen VL. 2021. Galectins in endothelial cell biology and angiogenesis: the basics. Biomolecules 11:1386.
- 48 Cheng YH, et al. 2022. Galectin-1 contributes to vascular remodeling and blood flow recovery after cerebral ischemia in mice. Transl Stroke Res. 13:160–170.
- 49 Boehm-Sturm P, et al. 2021. Phenotyping placental oxygenation in Lgals1 deficient mice using (19)F MRI. Sci Rep. 11:2126.
- 50 Barrientos G, et al. 2014. Involvement of galectin-1 in reproduction: past, present and future. Hum Reprod Update. 20:175–193.
- 51 Sharma S, Godbole G, Modi D. 2016. Decidual control of trophoblast invasion. Am J Reprod Immunol. 75:341–350.
- 52 Amaral LM, Wallace K, Owens M, LaMarca B. 2017. Pathophysiology and current clinical management of preeclampsia. Curr Hypertens Rep. 19:61.

- 53 Chen Q, et al. 2016. Evidence for differential glycosylation of trophoblast cell types. Mol Cell Proteomics. 15:1857–1866.
- 54 Dall'Olio F, Malagolini N, Chiricolo M, Trinchera M, Harduin-Lepers A. 2014. The expanding roles of the Sda/Cad carbohydrate antigen and its cognate glycosyltransferase B4GALNT2. Biochim Biophys Acta. 1840:443–453.
- 55 Li PT, Liao CJ, Yu LC, Wu WG, Chu ST. 2012. Localization of B4GALNT2 and its role in mouse embryo attachment. Fertil Steril. 97:1206–1212.e3.
- 56 Dell A, et al. 2003. Murine and human zona pellucida 3 derived from mouse eggs express identical O-glycans. Proc Natl Acad Sci U S A. 100:15631–15636.
- 57 Klisch K, et al. 2008. A tetraantennary glycan with bisecting N-acetylglucosamine and the Sd(a) antigen is the predominant N-glycan on bovine pregnancy-associated glycoproteins. Glycobiology 18:42–52.
- 58 Das SK, et al. 1994. Heparin-binding EGF-like growth factor gene is induced in the mouse uterus temporally by the blastocyst solely at the site of its apposition: a possible ligand for interaction with blastocyst EGF-receptor in implantation. *Development* 120: 1071–1083.

- 59 Xie H, et al. 2007. Maternal heparin-binding-EGF deficiency limits pregnancy success in mice. Proc Natl Acad Sci U S A. 104: 18315–18320.
- 60 Leach RE, et al. 2002. Pre-eclampsia and expression of heparinbinding EGF-like growth factor. *Lancet* 360:1215–1219.
- 61 Kuo PL, et al. 2012. Lung cancer-derived galectin-1 enhances tumorigenic potentiation of tumor-associated dendritic cells by expressing heparin-binding EGF-like growth factor. J Biol Chem. 287:9753–9764.
- 62 Theiler K. 1989. The house mouse: atlas of embryonic development. New York: Springer-Verlag, Havard.
- 63 Schwedhelm E, et al. 2007. High-throughput liquid chromatographic-tandem mass spectrometric determination of arginine and dimethylated arginine derivatives in human and mouse plasma. J Chromatogr B Analyt Technol Biomed Life Sci. 851: 211–219.
- 64 North SJ, et al. 2010. Mass spectrometric analysis of mutant mice. Methods Enzymol. 478:27–77.
- 65 Pennington KA, Schlitt JM, Schulz LC. 2012. Isolation of primary mouse trophoblast cells and trophoblast invasion assay. J Vis *Exp.* 59:e3202.